Author's response to reviews

Title: Hepatic inflammation mediated by hepatitis C virus core protein is ameliorated by blocking complement activation

Authors:

Ming-Ling Chang (kitty8210@yahoo.com)
Chau-Ting Yeh (chauting@adm.cgmh.org.tw)
Deng-Yn Lin (lindy0043@adm.cgmh.org.tw)
Yu-Pin Ho (ho12345678.tw@yahoo.com.tw)
Chen-Ming Hsu (hsu3060e@adm.cgmh.org.tw)
D. Montgomery Bissell (Montgomery.Bissell@ucsf.edu)

Version: 7 Date: 11 July 2009

Author's response to reviews:

Dear Editor,

The major revisions had been completed. Please check with the revised edition.

Best regards,

Ming-Ling Chang, M.D.
Liver Research Unit, Chang Gung memorial Hospital, Taoyuan, Taiwan